

XMAP215 promotes microtubule catastrophe by disrupting the growing microtubule end

Veronica Farmer, Göker Arpağ, Sarah Hall, and Marija Zanic

Corresponding Author(s): Marija Zanic, Vanderbilt University

Dovious Timelines	Submission Data:	2020 12 20
Review Timeline:	Submission Date.	2020-12-29
	Editorial Decision:	2021-02-25
	Revision Received:	2021-05-11
	Editorial Decision:	2021-06-30
	Revision Received:	2021-07-06

Monitoring Editor: Melissa Gardner

Scientific Editor: Andrea Marat

Transaction Report:

(Note: With the exception of the correction of typographical or spelling errors that could be a source of ambiguity, letters and reports are not edited. The original formatting of letters and referee reports may not be reflected in this compilation.)

DOI: https://doi.org/10.1083/jcb.202012144

February 25, 2021

Re: JCB manuscript #202012144

Dr. Marija Zanic Vanderbilt University Cell and Developmental Biology 465 21st Avenue South 4120 MRB3 Biosciences Building Nashville, TN 37232

Dear Dr. Zanic,

Thank you for submitting your manuscript entitled "XMAP215 promotes microtubule catastrophe by disrupting the growing microtubule end". The manuscript was assessed by expert reviewers, whose comments are appended to this letter. We invite you to submit a revision if you can address the reviewers' key concerns, as outlined here.

As you will see, the reviewers have differing opinions regarding the suitability of your study for JCB. I agree with reviewer #2 that there is a lack of mechanistic insight, however, an extended study to explore the mechanism for the phenomenon described here would require an article format, rather than a short report. As your study carefully identifies what is potentially an important observation - that the manner in which microtubules grow can ultimately influence their destruction, as defined by the probability of a catastrophe event, editorially I find your study suitable as a Report in JCB.

In revising, please address the reviewers comments as outlined:

Reviewer #1: The authors should carefully address all comments by reviewer #1.

Reviewer #2: The authors should also carefully address all comments by reviewer #2, and incorporate explanations and caveats as appropriate. In addition, please note the following regarding reviewer #2 comments:

- As noted above, I feel that the novelty and the level of mechanistic insight is appropriate for a report format.

- Given that the concentration of EB1 is kept constant for the experiments in the manuscript, I do not agree that effect of EB1 on microtubule end stability is problematic, as noted by reviewer #1.

- Reviewer #2 is also concerned about the synergism between EB1 and XMAP215 activities at the growing microtubule ends. This could be a concern if nonlinear biochemical effects were observed, for example in Fig. 3A. However, the response of EB1 comet length with increasing XMAP concentrations appears to be nearly linear in Fig. 3A, suggesting that the synergism between EB1 and XMAP215 may not be a nonlinear biochemical effect, but rather a result of changes in microtubule tip structure, as is explicitly explored in this manuscript. Therefore, I also disagree with the concern regarding synergism between EB1 and XMAP215.

- Reviewer #2 notes that the conclusions are vague. I do agree that the take-home message from this manuscript should be clarified. The authors note that "our results demonstrate that XMAP215 simultaneously promotes microtubule growth and catastrophe frequency without accelerating the GTP-hydrolysis rate, or otherwise decreasing the mean GTP-cap size". However, the data from Figures 4 provides insight into an explanation for how this may occur. A final cartoon to summarize the author's ideas regarding the mechanism to explain their results, along with a paragraph in the discussion, would help to clarify the take-home message for the paper, and to clarify what exactly the title "XMAP215 promotes microtubule catastrophe by disrupting the growing microtubule end" means.

- The authors may wish to rethink Fig. 5 in light of reviewer #2 comments - Lagree that the interpretation for Fig. 5 is unclear. One idea may be to eliminate the current Fig. 5, and replace it with a cartoon that demonstrates the take home message from the paper, based on Figs 1-4.

- Reviewer #2's minor points should all be addressed, with the exception of performing new experiments to remove the EB1 His-tag. This seems unnecessary given that the concentration of EB1 is held constant in the paper experiments. However, the authors may wish to comment on this issue.

Further suggestions:

Figure 2B and Figure 2C: Please add legends onto the graphs themselves, as it is difficult to search the legend to understand the graphs - eg, define dotted lines, different colored points, and fitted lines on the graphs themselves.

Figure 4B, right: perhaps avoid the acronym "SSR" which is not widely understood, and replace with complete text.

Figure 4D - It would be good to add p-values here, which can be calculated for proportions - one idea may be to report a p-value comparing the proportion of intact (full) comets vs disrupted (split+curled) comets, with and without XMAP, to provide a more definitive conclusion that XMAP is altering tip structure.

Figure 5 - As noted above by reviewer #2, Fig. 5 does not currently seem well connected to the remainder of the paper, and I would suggest dropping it. If it is not dropped, a clear connection between Fig. 4 and Fig. 5 should be described, as well as the contribution of Fig. 5 to the final conclusions in the paper.

New Figure: As noted above, a final cartoon to clarify the model put forward here, and to clarify the paper title, should be included.

While you are revising your manuscript, please also attend to the following editorial points to help expedite the publication of your manuscript. Please direct any editorial questions to the journal office.

GENERAL GUIDELINES:

Text limits: Character count for a Report is < 20,000, not including spaces. Count includes title page, abstract, introduction, results, discussion, acknowledgments, and figure legends. Count does not include materials and methods, references, tables, or supplemental legends.

Figures: Reports may have up to 5 main text figures. To avoid delays in production, figures must be prepared according to the policies outlined in our Instructions to Authors, under Data Presentation, https://jcb.rupress.org/site/misc/ifora.xhtml. All figures in accepted manuscripts will be screened prior to publication.

IMPORTANT: It is JCB policy that if requested, original data images must be made available. Failure to provide original images upon request will result in unavoidable delays in publication. Please ensure that you have access to all original microscopy and blot data images before submitting your revision.

Supplemental information: There are strict limits on the allowable amount of supplemental data. Reports may have up to 3 supplemental figures. Up to 10 supplemental videos or flash animations are allowed. A summary of all supplemental material should appear at the end of the Materials and methods section.

As you may know, the typical timeframe for revisions is three to four months. However, we at JCB realize that the implementation of social distancing and shelter in place measures that limit spread of COVID-19 also pose challenges to scientific researchers. Lab closures especially are preventing scientists from conducting experiments to further their research. Therefore, JCB has waived the revision time limit. We recommend that you reach out to the editors once your lab has reopened to decide on an appropriate time frame for resubmission. Please note that papers are generally considered through only one revision cycle, so any revised manuscript will likely be either accepted or rejected.

When submitting the revision, please include a cover letter addressing the reviewers' comments point by point. Please also highlight all changes in the text of the manuscript.

We hope that the comments below will prove constructive as your work progresses. We would be happy to discuss them further once you've had a chance to consider the points raised in this letter.

Thank you for this interesting contribution to Journal of Cell Biology. You can contact us at the journal office with any questions, cellbio@rockefeller.edu or call (212) 327-8588.

Sincerely,

Melissa Gardner Monitoring Editor

Andrea L. Marat Senior Scientific Editor

Journal of Cell Biology

Reviewer #1 (Comments to the Authors (Required)):

This is a very solid paper on the effects of the Microtubule polymerase XMAP215 on microtubule growth and catastrophe. The principal finding is that although XMAP215 increases the growth rate

and increases GTP cap size, it increases the catastrophe frequency. The interpretation, which is well supported by the data is that the microtubule lattice is defective when grown in XMAP. I thought the writing was clear and the presentation logical. I have some cosmetic comments.

1) The authors should be clearer about comet intensity. In some cases it is the intensity of the peak pixel if I understand it correctly, whereas in other places it is integrated intensity. It seems laid out pretty well in the methods, but the authors should be clearer in the results what they're referring to by intensity (perhaps rename peak pixel intensity or something like that).

2) Line 160: N=0?

3) The authors should better denote the different tubulin concentrations in Figure 3. For instance, in Fig 3B put tubulin on x-axis as well. As is, a casual reader might infer no change in growth rate from XMAP215 addition.

4) The authors should consider discussing Chretien's work "Mechanical Stress Induced Mechanism of MicrotubuleCatastrophes", J Mol Bio 2005. This is relevant for the catastrophe question.

Addendum to Review:

Reviewer 2 brings up the following point:

"The use of EB1 protein as an indicator of the GTP cap size is common in the field and may be appropriate in some cases. It is worth remembering, however, that EB1 has a very strong effect on microtubule end stability. This important fact will not be obvious to most readers, although this effect is clearly visible from carefully comparing Figs. 1B and S1B, which have vastly different scales. Moreover, as nicely shown by Zanic et al., 2013, there is a strong synergism between EB1 and XMAP215 activities at the growing microtubule ends. Thus, many of the conclusions in the current manuscript could be questioned because of this ill-understood synergism and the strong destabilizing effect by EB1."

Response to Reviewer 2:

I see that at 20 uM tubulin the catastrophe frequency is around 0.1/s in Fig 1B. In Fig S1B, which is at 20 uM tubulin, the zero XMAP215 value (yellow dotted line) is about 0.07/s. These seem to agree to me.

These experiments were carried out with 200 nM EB1-GFP in all conditions. In Zanic 2013, the Fig 1e catastrophe data used 5 uM tubulin and 400 nM EB1, and there was a clear enhancement of catastrophe frequency by EB1. The differences between the current and older data are worth pointing out by the authors, but I don't agree that these details undermine the study or make the current data questionable.

Reviewer #2 (Comments to the Authors (Required)):

Farmer and colleagues use in vitro assays to investigate XMAP215 activity at the growing microtubule end. Previous work, including by this group, found that XMAP215 is a strong polymerization factor that has little effect on catastrophe frequency. This seems counterintuitive because in a simplified model of microtubule stability, catastrophe frequency is determined by the size of the GTP cap, which is known to increase with increasing rate of microtubule polymerization.

However, it is also well established that other factors, including structure of the growing microtubule end, can also influence the frequency of microtubule catastrophe. Current work, which uses EB1 protein to visualize microtubule plus ends growing in the presence of XMAP215, reinforces these views. The authors report the XMAP215 increases complexity of the microtubule ends structure (they are often curled), resulting in elevated fluctuations in the microtubule growth rate and uncoupling of the correlation between the GTP cap size and catastrophe frequency. Understanding the mechanisms of microtubule dynamic instability and how it is controlled by MAPs is a very challenging and exciting area of cytoskeletal research. Despite my enthusiasm for this topic, and in general high guality of imaging and guantifications in the current study, I am concerned that its results do not provide significantly novel mechanistic insights that would be of interest to the general readership of JCB. First two figures in the paper establish important methodological background for this study, but the results reported in these figures are hardly novel. The rest of the figures provide more interesting contribution. However, the interpretation of these data is in my opinion problematic. The use of EB1 protein as an indicator of the GTP cap size is common in the field and may be appropriate in some cases. It is worth remembering, however, that EB1 has a very strong effect on microtubule end stability. This important fact will not be obvious to most readers, although this effect is clearly visible from carefully comparing Figs. 1B and S1B, which have vastly different scales. Moreover, as nicely shown by Zanic et al., 2013, there is a strong synergism between EB1 and XMAP215 activities at the growing microtubule ends. Thus, many of the conclusions in the current manuscript could be guestioned because of this ill-understood synergism and the strong destabilizing effect by EB1. There is not enough data to define contribution of these complicating factors on the main results (end morphology and rate fluctuations). Although it is well established by prior work and in the current manuscript that XMAP215 increases microtubule growth rate without suppressing catastrophe frequency even in the absence of EB1, the main findings described in this work could well be affected by EB1 presence.

Other significant criticism.

1) Some of the kymographs show unexplained peculiarities in microtubule dynamics.

Fig 1A. There is a very high rescue frequency for microtubule grown at low tubulin concentration, while with more EB1 this effect is gone. Generally, microtubules show little or no rescue at these tubulin concentrations (if highly competent tubulin is used). The high rescue frequency and its apparent dependence on tubulin concentration require additional examination and reasonable explanation

Fig 2A. Microtubule disassembly rate with no/low XMAP215 appears to be much lower than in other in vitro studies.

I am worried that there are some hidden (synergistic?) effects or technical deviations in this experimental system that need to be explained. Furthermore, these behaviors are very different from previous data by the same group (compare current Fig 1A with Fig 3B in Lawrence et al., 2018).

2) It is difficult to follow the authors' interpretations for findings illustrated with Fig. 5. It is known from prior work that microtubules grow much faster in the presence of XMAP215. Is it surprising that they undergo catastrophes at higher growth rates? The second conclusion drawn by the authors is that catastrophes take place with "more EB1". This is hardly evident from these data, which show almost similar levels of EB1 at catastrophe. More striking feature of the disassembling XMAP215 end is a strong drop in EB1 intensity over a relatively long time (20 s) prior to catastrophe. The authors should discuss these various features and seek mechanistic explanations. One concern is that the loss of EB1 could affect XMAP215 activity (through a loss of

synergistic interactions), leading to this unusual behavior, as already explained above.

More minor points:

3) Line 54: "early work demonstrated that increasing the microtubule growth rate by increasing the tubulin concentration in vitro is accompanied by a decrease in the catastrophe frequency (Walker et al., 1988)." This sentence does not do justice to a highly complex and controversial subject. Others found much smaller decrease (e.g. Gardner et al., 2011, Odde et al., 1995). It is also worth pointing out that Walker et al. findings at high tubulin concentration are based on a small data set, so referencing other similar studies would be appropriate.

4) Fig. 4. EB1 blobs and curling at the microtubule plus end are certainly indicative of some abnormal morphology, but why does this Figure title refer to these ends as "tapered" (which usually means gradual thinning of the tip)?

5) According to Zhu et al., 2009, His-tagged EB1, as used in the current work, has somewhat different properties than EB1 with no charged His extension. Have authors confirmed that the His-tag does not impact their results?

Rebuttal Letter

Farmer*, Arpag*, Hall & Zanic. XMAP215 promotes microtubule catastrophe by disrupting the growing microtubule end.

Response to the Editor

As you will see, the reviewers have differing opinions regarding the suitability of your study for JCB. I agree with reviewer #2 that there is a lack of mechanistic insight, however, an extended study to explore the mechanism for the phenomenon described here would require an article format, rather than a short report. As your study carefully identifies what is potentially an important observation - that the manner in which microtubules grow can ultimately influence their destruction, as defined by the probability of a catastrophe event, editorially I find your study suitable as a Report in JCB.

Thank you for your comments. We appreciate the suggestions and concerns raised by both reviewers and the editor, and have addressed them, as outlined below. Please note that in the revised manuscript document, we have highlighted the specific changes made as a response to the editor and reviewers in purple (also denoted by line numbers below).

In revising, please address the reviewers comments as outlined:

Reviewer #1: The authors should carefully address all comments by reviewer #1.

We have addressed all of Reviewer #1's comments, as detailed below.

Reviewer #2: The authors should also carefully address all comments by reviewer #2, and incorporate explanations and caveats as appropriate.

We have addressed all of Reviewer #2's comments, as detailed below.

In addition, please note the following regarding reviewer #2 comments:

- As noted above, I feel that the novelty and the level of mechanistic insight is appropriate for a report format.

- Given that the concentration of EB1 is kept constant for the experiments in the manuscript, I do not agree that effect of EB1 on microtubule end stability is problematic, as noted by reviewer #1.

- Reviewer #2 is also concerned about the synergism between EB1 and XMAP215 activities at the growing microtubule ends. This could be a concern if nonlinear biochemical effects were observed, for example in Fig. 3A. However, the response of EB1 comet length with increasing XMAP concentrations appears to be nearly linear in Fig. 3A, suggesting that the synergism between EB1 and XMAP215 may not be a nonlinear biochemical effect, but rather a result of changes in microtubule tip structure, as is explicitly explored in this manuscript. Therefore, I also disagree with the concern regarding synergism between EB1 and XMAP215.

We agree with the Editor on the above points and have provided a detailed response to these concerns raised by Reviewer #2 below.

- Reviewer #2 notes that the conclusions are vague. I do agree that the take-home message from this manuscript should be clarified. The authors note that "our results demonstrate that

XMAP215 simultaneously promotes microtubule growth and catastrophe frequency without accelerating the GTP-hydrolysis rate, or otherwise decreasing the mean GTP-cap size". However, the data from Figures 4 provides insight into an explanation for how this may occur. A final cartoon to summarize the author's ideas regarding the mechanism to explain their results, along with a paragraph in the discussion, would help to clarify the take-home message for the paper, and to clarify what exactly the title "XMAP215 promotes microtubule catastrophe by disrupting the growing microtubule end" means.

- The authors may wish to rethink Fig. 5 in light of reviewer #2 comments - I agree that the interpretation for Fig. 5 is unclear. One idea may be to eliminate the current Fig. 5, and replace it with a cartoon that demonstrates the take home message from the paper, based on Figs 1-4.

Based on the Reviewers' and Editor's comments, we realized that our motivation for previous Figure 5 (now Figure 4), interpretation of the presented results, and the proposed mechanism have not been sufficiently clear. To address this issue, we have now significantly revised previous Figure 5 (now Figure 4), added a cartoon schematic outlining our proposed model (new Figure 5), as well as clarified and elaborated our discussion, as suggested. We think that these revisions have significantly improved the clarity of our manuscript.

- Reviewer #2's minor points should all be addressed, with the exception of performing new experiments to remove the EB1 His-tag. This seems unnecessary given that the concentration of EB1 is held constant in the paper experiments. However, the authors may wish to comment on this issue.

We have addressed all of Reviewer #2's minor points, as detailed below.

Further suggestions:

Figure 2B and Figure 2C: Please add legends onto the graphs themselves, as it is difficult to search the legend to understand the graphs - eg, define dotted lines, different colored points, and fitted lines on the graphs themselves.

Please note that previous Figure 2 is now Figure 1; it has been revised as suggested.

Figure 4B, right: perhaps avoid the acronym "SSR" which is not widely understood, and replace with complete text.

Figure 4D - It would be good to add p-values here, which can be calculated for proportions - one idea may be to report a p-value comparing the proportion of intact (full) comets vs disrupted (split+curled) comets, with and without XMAP, to provide a more definitive conclusion that XMAP is altering tip structure.

Please note that previous Figure 4 is now Figure 3; it has been revised as suggested, and a p-value has been added in the caption, as well as the main text (line 202).

Figure 5 - As noted above by reviewer #2, Fig. 5 does not currently seem well connected to the remainder of the paper, and I would suggest dropping it. If it is not dropped, a clear connection between Fig. 4 and Fig. 5 should be described, as well as the contribution of Fig. 5 to the final conclusions in the paper.

Based on the reviewers' and editor's feedback, we realized that the point of the previous Figure 5 was not clear in the original version of the manuscript. We have now significantly revised

previous Figure 5 (now Figure 4), its description in the text, as well as clarified its contribution to the final conclusions of the manuscript (lines 206-235).

New Figure: As noted above, a final cartoon to clarify the model put forward here, and to clarify the paper title, should be included.

We have added the final cartoon as a new Figure 5, as suggested by the Editor.

Response to the Reviewers

Reviewer #1 (Comments to the Authors (Required)):

This is a very solid paper on the effects of the Microtubule polymerase XMAP215 on microtubule growth and catastrophe. The principal finding is that although XMAP215 increases the growth rate and increases GTP cap size, it increases the catastrophe frequency. The interpretation, which is well supported by the data is that the microtubule lattice is defective when grown in XMAP. I thought the writing was clear and the presentation logical. I have some cosmetic comments.

We thank the Reviewer for the constructive evaluation of our manuscript. We have addressed all of the Reviewer's comments, as summarized below.

1) The authors should be clearer about comet intensity. In some cases it is the intensity of the peak pixel if I understand it correctly, whereas in other places it is integrated intensity. It seems laid out pretty well in the methods, but the authors should be clearer in the results what they're referring to by intensity (perhaps rename peak pixel intensity or something like that).

Thank you for raising this important point. This comment inspired us to rethink our EB1 comet measurement approaches. We have now revised our methods for EB1 comet measurements, and significantly clarified what is being reported in each figure. Specifically, to allow direct comparison of EB1 comets between tubulin/EB1 and tubulin/EB1/XMAP215 conditions, we are now reporting EB1 comet decay lengths, instead of absolute intensities, in Figures S1 and 2. To determine EB1 comet decay lengths we adapted the previously-published approach from the Surrey lab (Bieling et al., 2007). While the absolute intensity measurements are highly dependent on imaging conditions (e.g. laser power, exposure time, TIRF angle), the EB1 comet decay length is an independent variable, which reflects the size of the GTP-cap, as set by the GTP hydrolysis rate and the microtubule growth rate. Thus, we think that the use of EB1 comet decay length is the most appropriate measurement for determining differences in the size of the GTP-cap between different experimental conditions. We now report EB1 intensity only at the moment of catastrophe in Figure 4, where the two experimental conditions (0 vs 3.13 nM XMAP215) are performed on the same day with the same imaging setup. In this case, our goal is to directly compare the relative amounts of EB1 remaining at the microtubule end at the onset of catastrophe, in order to test the hypothesis that there is a universal critical threshold of EB1 binding sites that needs to be reached to induce microtubule catastrophe (see Duellberg et al., 2016). Here, our methods adopt the previously-published approaches from the Surrey lab (Duellberg et al., 2016).

We have accordingly revised the figure legends, main text and the Methods to clarify our approaches.

2) Line 160: N=0?

The numbers originally reported in lines 159/160 referred to the total number of catastrophes observed in growth-rate-matching experiments in the absence vs. the presence of XMAP215 during the total time spent in growth for 20 microtubule kymographs analyzed for each condition We have updated the text accordingly to clarify this point, now reporting both the number of catastrophes observed, as well as the number of growth events analyzed (lines 162-164).

3) The authors should better denote the different tubulin concentrations in Figure 3. For instance, in Fig 3B put tubulin on x-axis as well. As is, a casual reader might infer no change in growth rate from XMAP215 addition.

Thank you for this suggestion. We have updated the figure (now Figure 2) to reflect tubulin concentrations, as suggested by the Reviewer.

4) The authors should consider discussing Chretien's work "Mechanical Stress Induced Mechanism of MicrotubuleCatastrophes", J Mol Bio 2005. This is relevant for the catastrophe question.

Thank you for pointing out this omission. We have now included the relevant references in our revised Conclusions section. Specifically, we have added the following text (lines 270-275):

"...Notably, a variety of growing-end configurations have been observed by structural studies (Mcintosh et al., 2018)(Gudimchuk et al., 2020)(Chrétien et al., 1995)(Guesdon et al., 2016)(Atherton et al., 2018) (Mandelkow et al., 1991)(Reid et al., 2019), and it has been previously proposed that some of these end configurations may indeed be energetically unfavorable, leading to microtubule catastrophe (Chrétien and Fuller, 2000; Hunyadi et al., 2005)..."

Addendum to Review:

Reviewer 2 brings up the following point:

"The use of EB1 protein as an indicator of the GTP cap size is common in the field and may be appropriate in some cases. It is worth remembering, however, that EB1 has a very strong effect on microtubule end stability. This important fact will not be obvious to most readers, although this effect is clearly visible from carefully comparing Figs. 1B and S1B, which have vastly different scales. Moreover, as nicely shown by Zanic et al., 2013, there is a strong synergism between EB1 and XMAP215 activities at the growing microtubule ends. Thus, many of the conclusions in the current manuscript could be questioned because of this ill-understood synergism and the strong destabilizing effect by EB1."

Response to Reviewer 2:

I see that at 20 uM tubulin the catastrophe frequency is around 0.1/s in Fig 1B. In Fig S1B, which is at 20 uM tubulin, the zero XMAP215 value (yellow dotted line) is about 0.07/s. These seem to agree to me.

These experiments were carried out with 200 nM EB1-GFP in all conditions. In Zanic 2013, the Fig 1e catastrophe data used 5 uM tubulin and 400 nM EB1, and there was a clear enhancement of catastrophe frequency by EB1. The differences between the current and older data are worth pointing out by the authors, but I don't agree that these details undermine the study or make the current data questionable.

Thank you for this Addendum. We have addressed all of the Reviewer 2's comments and concerns, as detailed below.

Reviewer #2 (Comments to the Authors (Required)):

Farmer and colleagues use in vitro assays to investigate XMAP215 activity at the growing microtubule end. Previous work, including by this group, found that XMAP215 is a strong polymerization factor that has little effect on catastrophe frequency. This seems counterintuitive because in a simplified model of microtubule stability, catastrophe frequency is determined by the size of the GTP cap, which is known to increase with increasing rate of microtubule polymerization. However, it is also well established that other factors, including structure of the growing microtubule end, can also influence the frequency of microtubule catastrophe. Current work, which uses EB1 protein to visualize microtubule plus ends growing in the presence of XMAP215, reinforces these views. The authors report the XMAP215 increases complexity of the microtubule ends structure (they are often curled), resulting in elevated fluctuations in the microtubule growth rate and uncoupling of the correlation between the GTP cap size and catastrophe frequency.

Understanding the mechanisms of microtubule dynamic instability and how it is controlled by MAPs is a very challenging and exciting area of cytoskeletal research. Despite my enthusiasm for this topic, and in general high quality of imaging and quantifications in the current study, I am concerned that its results do not provide significantly novel mechanistic insights that would be of interest to the general readership of JCB.

We thank the Reviewer for the critical evaluation of our manuscript. We have addressed the raised concerns point-by-point, as outlined below.

First two figures in the paper establish important methodological background for this study, but the results reported in these figures are hardly novel. The rest of the figures provide more interesting contribution. However, the interpretation of these data is in my opinion problematic. The use of EB1 protein as an indicator of the GTP cap size is common in the field and may be appropriate in some cases. It is worth remembering, however, that EB1 has a very strong effect on microtubule end stability. This important fact will not be obvious to most readers, although this effect is clearly visible from carefully comparing Figs. 1B and S1B, which have vastly different scales. Moreover, as nicely shown by Zanic et al., 2013, there is a strong synergism between EB1 and XMAP215 activities at the growing microtubule ends. Thus, many of the conclusions in the current manuscript could be questioned because of this ill-understood synergism and the strong destabilizing effect by EB1. There is not enough data to define contribution of these complicating factors on the main results (end morphology and rate fluctuations). Although it is well established by prior work and in the current manuscript that XMAP215 increases microtubule growth rate without suppressing catastrophe frequency even in the absence of EB1, the main findings described in this work could well be affected by EB1 presence.

We agree with the reviewer that many previous studies have reported the dependence of microtubule growth rate and microtubule catastrophe frequency on tubulin concentration *in vitro* (see e.g. O'Brien et al., 1990, Drechsel et al., 1992, Odde et al., 1995, Gardner et al., 2011, Piedra et al., 2016, Chaaban et al., 2018, Strothman et al., 2019, Arpağ et al., 2020). Additionally, the size of the EB1/Mal3 comets was previously shown to increase with the microtubule growth rate when achieved by increasing tubulin concentrations (Bieling et al., 2007)(Duellberg et al., 2016)(Strothman et al., 2019). However, the relationship between microtubule catastrophe frequency and the EB1 comet size over a range of tubulin concentrations has, to our knowledge, never been reported previously. Our results in the previous Figure 1 (now Figure S1) for the first time directly establish the canonical relationship between the GTP-cap size (as determined by the EB1 comet length) and the microtubule catastrophe frequency, in conditions where the tubulin concentration is modulated. This

relationship presents the basis of the standard model of dynamic instability (i.e. larger GTP-cap leads to prolonged microtubule lifetime), which is subsequently challenged by our observations in the presence of XMAP215, and thus we think that it warrants an independent figure. In the revised manuscript, we present this result in the supplemental Figure S1. This restructuring allowed us to add a model cartoon as a new Figure 5, as suggested by the Editor.

Our results presented in the original Figure 2 (now Figure 1) demonstrate that the addition of XMAP215 increases microtubule catastrophe frequency, in spite of its well-known effect of strongly increasing the microtubule growth rate. Although simultaneous promotion of growth and catastrophe has been previously observed in the presence of XMAP215 and EB1 (Zanic et al., 2013), the increase in catastrophe was primarily attributed to the effects of EB1, which is indeed a known catastrophe factor in vitro. Here, we perform a full titration in XMAP215 concentration in the background of fixed EB1 concentration and show that the addition of XMAP215 on its own causes an increase in microtubule catastrophe frequency. We demonstrate that XMAP215driven increase in catastrophe frequency is observed both in the presence (now Figure 1) and in the absence of EB1 (now Figure S2). Importantly, unlike the synergistic effect of XMAP215 and EB1 on microtubule growth (where the combined effects of the two proteins greatly exceed the product of their individual effects), the extent of the catastrophe frequency increase by XMAP215 in the presence of EB1 (~ 2-fold, Figure 1) is not larger than the XMAP215-induced increase in catastrophe frequency in the absence of EB1 (2 to 4-fold, Figure S2). Taken together, these results demonstrate that the observed effect of XMAP215 on catastrophe frequency is a direct consequence of XMAP215 alone, rather than a synergistic effect dependent on EB1. Moreover, the observed functional dependence of catastrophe frequency on XMAP215 concentration (both in the presence and in the absence of EB1) is clearly distinct from the dose-dependent effects of XMAP215 on microtubule growth rate, suggesting that the mechanisms driving the observed XMAP215-mediated effects on microtubule catastrophe are distinct from those driving the growth acceleration (both alone and in synergy with EB1).

Other significant criticism.

1) Some of the kymographs show unexplained peculiarities in microtubule dynamics. Fig 1A. There is a very high rescue frequency for microtubule grown at low tubulin concentration, while with more EB1 this effect is gone. Generally, microtubules show little or no rescue at these tubulin concentrations (if highly competent tubulin is used). The high rescue frequency and its apparent dependence on tubulin concentration require additional examination and reasonable explanation.

Thank you for raising this concern. Our previous analysis had focused solely on microtubule catastrophe and growth rates, as this is the subject of the manuscript. We have now additionally measured microtubule rescue frequencies and shrinkage rates in our experiments, to compare with previously-published data. In addition to the tubulin concentration, it is well known that buffer conditions, including KCI and glycerol concentrations, significantly impact microtubule dynamics (see e.g. Wieczorek et al., 2013, Schilstra et al., 1991, O'Brien et al., 1990, Stewart et al., 1990). To that end, please note that our buffer conditions for tubulin titration in previous Figure 1 (now Figure S1) included 17 mM KCI and 2.5% glycerol (to adequately maintain the composition of the XMAP215 storage buffer, even in conditions where XMAP215 was not used), which is significantly different from e.g. Lawrence et al., 2018. In Lawrence et al., 2018, where 50-100 mM KCI and no glycerol were used, rescue frequency was ~ 1 event/minute of shrinkage at 8 μ M tubulin. Here, for 12 μ M tubulin condition growing at 17 nm/s (the lowest concentration used in this study), our measured rescue frequency is 3.9 \pm 0.8 events/minute of shrinkage (total of 26 rescues observed over 20 microtubule kymographs, 300 minutes of observed dynamics, of which 400 seconds spent in shrinkage). Note that the classic Walker et

al., 1988 study reported plus-end rescue frequencies in the range of 2-5 min⁻¹ over a range of 7-16 μ M tubulin. Similarly, our recent study (Arpağ et al., 2020) reported plus-end rescue frequencies in the range of 2-8 min⁻¹ over a range of 5-12 μ M tubulin. Thus, our rescue frequency measured here, even with the distinct buffer composition, is consistent with previously published studies.

Furthermore, please note that our tubulin titration in previous Figure 1 (now Figure S1) was performed in the background of a constant EB1 concentration (200 nM EB1), so the data do not represent any EB1-concentration-dependent effects on rescue frequency. As far as the rescue dependence on tubulin concentration is concerned, our titration used relatively high concentrations of tubulin, and rescue frequencies become increasingly difficult to measure at tubulin concentrations of 20 μ M and above because at these tubulin concentrations very few catastrophe events (and thus possibilities for rescues) are observed. We did manage to measure the rescue frequency in our 20 μ M tubulin condition and determined it to be 4.2 ± 1.6 min⁻¹ (total of 7 rescues observed over 20 microtubule kymographs, 300 minutes of observed dynamics, of which 101 seconds spent in shrinkage), which is not significantly different from our 12 μ M tubulin measurement. Based on these limited data, we cannot make any meaningful comments on tubulin-concentration dependence of rescue frequency, but we do note that our data are consistent with previous studies showing no obvious dependence of plus-end rescue frequency on tubulin concentration (see e.g. Walker et al., 1988, Fees and Moore, 2019, Arpağ et al., 2020).

Notably, in light of our new rescue frequency measurements, we realized that one of the kymographs originally shown in Figure 1A (now Figure S1A) was not representative with respect to the rescue frequency, and have thus replaced it with a kymograph representative of all dynamics parameters measured in this condition.

Finally, we are also including here an overloaded SDS-PAGE gel of our purified tubulin showing no significant bands other than the tubulin itself.



SDS-Page gel stained with coomassie. 200 µg of purified tubulin loaded

Fig 2A. Microtubule disassembly rate with no/low XMAP215 appears to be much lower than in other in vitro studies.

I am worried that there are some hidden (synergistic?) effects or technical deviations in this experimental system that need to be explained. Furthermore, these behaviors are very different from previous data by the same group (compare current Fig 1A with Fig 3B in Lawrence et al., 2018).

Please note that the imaging conditions in previous Figure 2A (now Figure 1A) used an acquisition rate of 5 frames per second (fps), which is significantly (25x) faster than the acquisition rate used in Figure 3 of Lawrence et al., 2018 (0.2 fps). Consequently, the aspect ratio of the kymograph is very different, clearly showing the full duration of the microtubule shrinkage occurring over several imaging frames, unlike in Figure 3 of Lawrence et al., 2018, as well as most other studies, where slow acquisition rate (typically 0.2 fps = 1 frame every 5 seconds) does not allow time-resolved imaging of microtubule shrinkage. To clarify this point, we have now added additional details of the imaging conditions in the Methods. Furthermore, we have now also measured the microtubule shrinkage rates for our experiments shown in previous Figure 2A (now Figure 1A), and found the mean shrinkage rates to be 230 ± 40 nm/s (SD, N=6 independent experimental repeats, 739 total shrinkage events analyzed) in the absence of XMAP215, and 370 ± 80 nm/s (SD, N=3 independent experimental repeats, 656 total shrinkage events analyzed) in the presence of low XMAP215 concentration (6.25 nM XMAP215). These rates are consistent with the previously-published microtubule shrinkage rates (see e.g. Lawrence et al., 2018, Walker et al., 1988, Arpağ et al., 2020). The observed increase in shrinkage rate with XMAP215 is also consistent with previous reports of XMAP215's effects on microtubule shrinkage (Vasguez et al., 1994)(Zanic et al., 2013). Thus, we conclude that our data show no deviations from the expected microtubule disassembly rates, as based on the previous literature.

2) It is difficult to follow the authors' interpretations for findings illustrated with Fig. 5. It is known from prior work that microtubules grow much faster in the presence of XMAP215. Is it surprising that they undergo catastrophes at higher growth rates? The second conclusion drawn by the authors is that catastrophes take place with "more EB1". This is hardly evident from these data, which show almost similar levels of EB1 at catastrophe. More striking feature of the disassembling XMAP215 end is a strong drop in EB1 intensity over a relatively long time (20 s) prior to catastrophe. The authors should discuss these various features and seek mechanistic explanations. One concern is that the loss of EB1 could affect XMAP215 activity (through a loss of synergistic interactions), leading to this unusual behavior, as already explained above.

Based on the Reviewers' and Editor's comments, we realized that our original motivation for and interpretation of the results in the original Figure 5 (now Figure 4) was not clear. The goal of this figure is to test the hypothesis that a specific threshold in the GTP-cap size needs to be reached as a prerequisite for catastrophe. The Reviewer is absolutely correct that microtubules will grow much faster in the presence of XMAP215. In the absence of XMAP215, it has been shown that microtubules undergo a significant slowdown period prior to catastrophe (see e.g. Maurer et al., 2014). This slowdown allows for the decay of the GTP-cap, and it has been proposed that a minimal cap density is required for microtubule stability (Duellberg et al., 2016). This GTP-cap threshold was shown to be independent of the growth rate measured during the microtubule growth phase significantly prior to the onset of catastrophe. Thus, regardless of how fast the microtubule initially grew, once the GTP-cap decay crosses the minimal threshold, catastrophe ensues (Duellberg et al., 2016). Our results show that the presence of XMAP215 changes this process, such that a microtubule can undergo catastrophe without having to slow down to the level necessary in the absence of XMAP215. Furthermore, we show that catastrophe in the

presence of XMAP215 occurs with significantly more EB1 sites still occupied, when compared to the EB1 levels observed in the absence of XMAP215. Thus, our results demonstrate that the presence of XMAP215 destabilizes the microtubule end and increases the minimal cap threshold needed for catastrophe.

To clarify these findings, we have now modified the previous Figure 5 (now Figure 4), as well as significantly clarified the accompanying text. Specifically, we have focused our figure on the last 10 seconds of microtubule growth prior to catastrophe, to draw attention that we are comparing instantaneous growth rates at catastrophe (i.e. measured only during the last 1 second prior to the catastrophe onset). We are now analyzing and showing traces obtained at the lowest XMAP215 concentration (3.13 nM), at which the effects of XMAP215 on growth rate acceleration are minimal, while the effects on catastrophe frequency have already reached their full magnitude (please see current Figure 1). As we originally demonstrated with higher XMAP215 concentrations, even with the lowest XMAP215 concentration we see that the instantaneous growth rate and the EB1 levels at the time of catastrophe are significantly larger in the absence of XMAP215 (p <0.001 for instantaneous growth rate, and p = 0.003 for EB1 levels, unpaired Welch's t-test). We think that the ability of XMAP215 to induce microtubule catastrophe with a larger GTP-cap provides a significant insight into its molecular mechanism of action. To further illustrate our proposed model, we have now included a cartoon schematic as a new Figure 5.

More minor points:

3) Line 54: "early work demonstrated that increasing the microtubule growth rate by increasing the tubulin concentration in vitro is accompanied by a decrease in the catastrophe frequency (Walker et al., 1988)." This sentence does not do justice to a highly complex and controversial subject. Others found much smaller decrease (e.g. Gardner et al., 2011, Odde et al., 1995). It is also worth pointing out that Walker et al. findings at high tubulin concentration are based on a small data set, so referencing other similar studies would be appropriate.

Thank you for pointing this out; we very much appreciate the complexity of the subject. We have now added additional text and references to address this point (lines 53-58):

"Although the exact functional dependence and the extent of catastrophe suppression with increasing tubulin concentration have varied between subsequent reports, the finding that increase in tubulin concentration correlates with a decrease in catastrophe frequency remains generally supported (O'Brien et al., 1990)(Drechsel et al., 1992)(Odde et al., 1995)(Gardner et al., 2011)(Bowne-Anderson et al., 2013)(Piedra et al., 2016) (Chaaban et al., 2018)(Strothman et al., 2019)(Arpağ et al., 2020)..."

4) Fig. 4. EB1 blobs and curling at the microtubule plus end are certainly indicative of some abnormal morphology, but why does this Figure title refer to these ends as "tapered" (which usually means gradual thinning of the tip)?

We appreciate this concern and have put a lot of thought into what would be the most appropriate terminology for our observations. Our TIRF data demonstrate subsets of protofilaments protruding ahead of the core microtubule lattice to produce split and/or curled EB1 comets. This phenomenon, where some protofilaments protrude ahead of others has been widely referred to as 'tapered' ever since the early cryo-EM observations by Mandelkow et al., 1991 (see e.g. Aher and Akhmanova, 2018 for a more extensive review on this topic). Therefore, we think that the use of the term 'tapered' is most consistent with the existing literature on the topic. 5) According to Zhu et al., 2009, His-tagged EB1, as used in the current work, has somewhat different properties than EB1 with no charged His extension. Have authors confirmed that the His-tag does not impact their results?

We are aware of the Zhu et al., 2009 study, which suggested potential concentration-dependent effects of His-tagged EB1 proteins in bulk assays using very high EB1 concentrations (2-20 μ M). In our study, we use a constant concentration of EB1-GFP (200 nM) across all conditions involving EB1. The main goal of our study is to investigate the effects of XMAP215 on microtubule catastrophe, and our data demonstrate that these do not depend on the presence of EB1. Overall, we do not think that studying the potential effects of His-tag on EB1 will contribute to the results of our study, as agreed by the editor.

REFERENCES

- Aher, A., and A. Akhmanova. 2018. Tipping microtubule dynamics, one protofilament at a time. *Curr. Opin. Cell Biol.* 50:86–93. doi:10.1016/j.ceb.2018.02.015.
- Arpağ, G., E.J. Lawrence, V.J. Farmer, S.L. Hall, and M. Zanic. 2020. Collective effects of XMAP215, EB1, CLASP2, and MCAK lead to robust microtubule treadmilling. *PNAS*. 117:12847–12855. doi:10.1073/pnas.2003191117.
- Atherton, J., M. Stouffer, F. Francis, and C.A. Moores. 2018. Microtubule architecture in vitro and in cells revealed by cryo-electron tomography. *Acta Crystallogr. Sect. D Struct. Biol.* 74:1–13. doi:10.1107/S2059798318001948.
- Bieling, P., L. Laan, H. Schek, E.L. Munteanu, L. Sandblad, M. Dogterom, D. Brunner, and T. Surrey. 2007. Reconstitution of a microtubule plus-end tracking system in vitro. *Nature*. 450:1100–1105. doi:10.1038/nature06386.
- Bowne-Anderson, H., M. Zanic, M. Kauer, and J. Howard. 2013. Microtubule dynamic instability: A new model with coupled GTP hydrolysis and multistep catastrophe. *BioEssays*. 35:452–461. doi:10.1002/bies.201200131.
- Chaaban, S., S. Jariwala, C.T. Hsu, S. Redemann, J.M. Kollman, T. Müller-Reichert, D. Sept, K.H. Bui, and G.J. Brouhard. 2018. The Structure and Dynamics of C. elegans Tubulin Reveals the Mechanistic Basis of Microtubule Growth. *Dev. Cell*. 47:191-204.e8. doi:10.1016/j.devcel.2018.08.023.
- Chrétien, D., and S.D. Fuller. 2000. Microtubules switch occasionally into unfavorable configurations during elongation. *J. Mol. Biol.* 298:663–676. doi:10.1006/jmbi.2000.3696.
- Chrétien, D., S.D. Fuller, and E. Karsenti. 1995. Structure of growing microtubule ends: Twodimensional sheets close into tubes at variable rates. *J. Cell Biol.* 129:1311–1328. doi:10.1083/jcb.129.5.1311.
- Drechsel, D.N., A.A. Hyman, M.H. Cobb, and M.W. Kirschner. 1992. Modulation of the dynamic instability of tubulin assembly by the microtubule-associated protein tau. *Mol. Biol. Cell*. 3:1141–54. doi:10.1091/mbc.3.10.1141.
- Duellberg, C., N.I. Cade, D. Holmes, and T. Surrey. 2016. The size of the EB cap determines instantaneous microtubule stability. *Elife*. 5:1–23. doi:10.7554/eLife.13470.
- Fees, C.P., and J.K. Moore. 2019. A unified model for microtubule rescue. *Mol. Biol. Cell.* doi:10.1091/mbc.E18-08-0541.
- Gardner, M.K., M. Zanic, C. Gell, V. Bormuth, and J. Howard. 2011. Depolymerizing kinesins Kip3 and MCAK shape cellular microtubule architecture by differential control of catastrophe. *Cell*. 147:1092–1103. doi:10.1016/j.cell.2011.10.037.
- Gudimchuk, N.B., E. V Ulyanov, E.O. Toole, C.L. Page, D.S. Vinogradov, G. Morgan, G. Li, J.K. Moore, E. Szczesna, A. Roll-mecak, F.I. Ataullakhanov, and J.R. Mcintosh. 2020.
 Mechanisms of microtubule dynamics and force generation examined with computational modeling and electron cryotomography. *Nat. Commun.* 1–15. doi:10.1038/s41467-020-17553-2.
- Guesdon, A., F. Bazile, R.M. Buey, R. Mohan, S. Monier, R.R. García, M. Angevin, C. Heichette, R. Wieneke, R. Tampé, L. Duchesne, A. Akhmanova, M.O. Steinmetz, and D. Chrétien. 2016. EB1 interacts with outwardly curved and straight regions of the microtubule lattice. *Nat. Cell Biol.* 1. doi:10.1038/ncb3412.
- Hunyadi, V., D. Chrétien, and I.M. Jánosi. 2005. Mechanical stress induced mechanism of microtubule catastrophes. *J. Mol. Biol.* 348:927–938. doi:10.1016/j.jmb.2005.03.019.
- Lawrence, E.J., G. Arpag, S.R. Norris, and M. Zanic. 2018. Human CLASP2 specifically regulates microtubule catastrophe and rescue. *Mol. Biol. Cell*. 29:1168–1177. doi:10.1091/mbc.E18-01-0016.
- Mandelkow, E.-M., E. Mandelkow, and R.A. Milliganll. 1991. Microtubule dynamics and microtubule caps: a time-resolved cryo- electron microscopy study. *J. Cell Biol.* 114:977–

991. doi:10.1083/jcb.114.5.977.

- Maurer, S.P., N.I. Cade, G. Bohner, N. Gustafsson, E. Boutant, and T. Surrey. 2014. EB1 accelerates two conformational transitions important for microtubule maturation and dynamics. *Curr. Biol.* 24:372–384. doi:10.1016/j.cub.2013.12.042.
- Mcintosh, J.R., E.O. Toole, G. Morgan, J. Austin, E. Ulyanov, F.I. Ataullakhanov, and N.B. Gudimchuk. 2018. Microtubules grow by the addition of bent guanosine triphosphate tubulin to the tips of curved protofilaments. *J. Cell Biol.* 1–25.
- O'Brien, E.T., E.D. Salmon, R.A. Walker, and H.P. Erickson. 1990. Effects of Magnesium on the Dynamic Instability of Individual Microtubules. *Biochemistry*. 29:6648–6656. doi:10.1021/bi00480a014.
- Odde, D.J., L. Cassimeris, and H.M. Buettner. 1995. Kinetics of microtubule catastrophe assessed by probabilistic analysis. *Biophys. J.* 69:796–802. doi:10.1016/S0006-3495(95)79953-2.
- Piedra, F.-A., T. Kim, E.S. Garza, E.A. Geyer, A. Burns, X. Ye, L.M. Rice, and S.L. Reck-Peterson. 2016. GDP-to-GTP exchange on the microtubule end can contribute to the frequency of catastrophe. *Mol. Biol. Cell*. 27:3515–3525. doi:10.1091/mbc.E16-03-0199.
- Reid, T.A., C. Coombes, S. Mukherjee, R.R. Goldblum, K. White, S. Parmar, M. Mcclellan, M. Zanic, N. Courtemanche, and M.K. Gardner. 2019. Structural state recognition facilitates tip tracking of EB1 at growing microtubule ends. *Elife*. 1–32. doi:10.1101/636092.
- Schilstra, M.J., P.M. Bayley, and S.R. Martin. 1991. The effect of solution composition on microtubule dynamic instability. *Biochem. J.* 277:839–847.
- Stewart, R.J., K.W. Farrell, and L. Wilson. 1990. Role of GTP Hydrolysis in Microtubule Polymerization: Evidence for a Coupled Hydrolysis Mechanism. *Biochemistry*. 29:6489– 6498. doi:10.1021/bi00479a022.
- Strothman, C., V. Farmer, G. Arpağ, N. Rodgers, M. Podolski, S. Norris, R. Ohi, and M. Zanic. 2019. Microtubule minus-end stability is dictated by the tubulin off-rate. *J. Cell Biol.* 218:2841–2853. doi:10.1083/jcb.201905019.
- Vasquez, R.J., D.L. Gard, and L. Cassimeris. 1994. XMAP from Xenopus Eggs Promotes Rapid Plus End Assembly of Microtubules and Rapid Microtubule Polymer Turnover. J. Cell Biol. 127:985–993.
- Walker, R.A., E.T. O'Brien, N.K. Pryer, M.F. Soboeiro, W.A. Voter, H.P. Erickson, and E.D. Salmon. 1988. Dynamic Instability of Individual Microtubules. *J. Cell Biol.* 107:1437–1448.
- Wieczorek, M., S. Chaaban, and G.J. Brouhard. 2013. Macromolecular crowding pushes catalyzed microtubule growth to near the theoretical limit. *Cell. Mol. Bioeng.* 6:383–392. doi:10.1007/s12195-013-0292-9.
- Zanic, M., P.O. Widlund, A.A. Hyman, and J. Howard. 2013. Synergy between XMAP215 and EB1 increases microtubule growth rates to physiological levels. *Nat. Cell Biol.* 15:688–693. doi:10.1038/ncb2744.
- Zhu, Z.C., K.K. Gupta, A.R. Slabbekoorn, B.A. Paulson, E.S. Folker, and H. V. Goodson. 2009. Interactions between EB1 and microtubules: Dramatic effect of affinity tags and evidence for cooperative behavior. *J. Biol. Chem.* 284:32651–32661. doi:10.1074/jbc.M109.013466.

June 30, 2021

RE: JCB Manuscript #202012144R

Dr. Marija Zanic Vanderbilt University Cell and Developmental Biology 465 21st Avenue South 4120 MRB3 Biosciences Building Nashville, TN 37232

Dear Dr. Zanic:

Thank you for submitting your revised manuscript entitled "XMAP215 promotes microtubule catastrophe by disrupting the growing microtubule end". Reviewer #1 now supports publication, and while reviewer #2 was not available to re-review I have carefully read your responses to both reviewers and agree that you did an excellent job with the revisions. Therefore, we would be happy to publish your paper in JCB pending final revisions necessary to meet our formatting guidelines (see details below).

To avoid unnecessary delays in the acceptance and publication of your paper, please read the following information carefully.

A. MANUSCRIPT ORGANIZATION AND FORMATTING:

Full guidelines are available on our Instructions for Authors page, https://jcb.rupress.org/submissionguidelines#revised. **Submission of a paper that does not conform to JCB guidelines will delay the acceptance of your manuscript.**

1) Text limits: Character count for Reports is < 20,000, not including spaces. Count includes title page, abstract, introduction, combined results and discussion, acknowledgments, and figure legends. Count does not include materials and methods, references, tables, or supplemental legends.

2) Figures limits: Reports may have up to 5 main text figures.

3) Figure formatting: Scale bars must be present on all microscopy images, including inset magnifications. Molecular weight or nucleic acid size markers must be included on all gel electrophoresis.

4) Statistical analysis: Error bars on graphic representations of numerical data must be clearly described in the figure legend. The number of independent data points (n) represented in a graph must be indicated in the legend. Statistical methods should be explained in full in the materials and methods. For figures presenting pooled data the statistical measure should be defined in the figure legends. Please also be sure to indicate the statistical tests used in each of your experiments (either in the figure legend itself or in a separate methods section) as well as the parameters of the test (for example, if you ran a t-test, please indicate if it was one- or two-sided, etc.). Also, if you

used parametric tests, please indicate if the data distribution was tested for normality (and if so, how). If not, you must state something to the effect that "Data distribution was assumed to be normal but this was not formally tested."

5) Abstract and title: The abstract should be no longer than 160 words and should communicate the significance of the paper for a general audience. The title should be less than 100 characters including spaces. Make the title concise but accessible to a general readership.

6) Materials and methods: Should be comprehensive and not simply reference a previous publication for details on how an experiment was performed. Please provide full descriptions in the text for readers who may not have access to referenced manuscripts.

7) Please be sure to provide the sequences for all of your primers/oligos and RNAi constructs in the materials and methods. You must also indicate in the methods the source, species, and catalog numbers (where appropriate) for all of your antibodies. Please also indicate the acquisition and quantification methods for immunoblotting/western blots.

8) Microscope image acquisition: The following information must be provided about the acquisition and processing of images:

- a. Make and model of microscope
- b. Type, magnification, and numerical aperture of the objective lenses
- c. Temperature
- d. Imaging medium
- e. Fluorochromes
- f. Camera make and model
- g. Acquisition software

h. Any software used for image processing subsequent to data acquisition. Please include details and types of operations involved (e.g., type of deconvolution, 3D reconstitutions, surface or volume rendering, gamma adjustments, etc.).

9) References: There is no limit to the number of references cited in a manuscript. References should be cited parenthetically in the text by author and year of publication. Abbreviate the names of journals according to PubMed.

10) Supplemental materials: There are strict limits on the allowable amount of supplemental data. Reports may have up to 3 supplemental display items (figures and tables). Please also note that tables, like figures, should be provided as individual, editable files. A summary of all supplemental material should appear at the end of the Materials and methods section.

11) eTOC summary: A ~40-50-word summary that describes the context and significance of the findings for a general readership should be included on the title page. The statement should be written in the present tense and refer to the work in the third person.

12) Conflict of interest statement: JCB requires inclusion of a statement in the acknowledgements regarding competing financial interests. If no competing financial interests exist, please include the following statement: "The authors declare no competing financial interests." If competing interests are declared, please follow your statement of these competing interests with the following statement: "The authors declare no further competing financial interests."

13) ORCID IDs: ORCID IDs are unique identifiers allowing researchers to create a record of their

various scholarly contributions in a single place. At resubmission of your final files, please consider providing an ORCID ID for as many contributing authors as possible.

14) A separate author contribution section following the Acknowledgments. All authors should be mentioned and designated by their full names. We encourage use of the CRediT nomenclature.

B. FINAL FILES:

Please upload the following materials to our online submission system. These items are required prior to acceptance. If you have any questions, contact JCB's Managing Editor, Lindsey Hollander (Ihollander@rockefeller.edu).

-- An editable version of the final text (.DOC or .DOCX) is needed for copyediting (no PDFs).

-- High-resolution figure and MP4 video files: See our detailed guidelines for preparing your production-ready images, https://jcb.rupress.org/fig-vid-guidelines.

-- Cover images: If you have any striking images related to this story, we would be happy to consider them for inclusion on the journal cover. Submitted images may also be chosen for highlighting on the journal table of contents or JCB homepage carousel. Images should be uploaded as TIFF or EPS files and must be at least 300 dpi resolution.

It is JCB policy that if requested, original data images must be made available to the editors. Failure to provide original images upon request will result in unavoidable delays in publication. Please ensure that you have access to all original data images prior to final submission.

The license to publish form must be signed before your manuscript can be sent to production. A link to the electronic license to publish form will be sent to the corresponding author only. Please take a moment to check your funder requirements before choosing the appropriate license.

Thank you for your attention to these final processing requirements. Please revise and format the manuscript and upload materials within 7 days. If complications arising from measures taken to prevent the spread of COVID-19 will prevent you from meeting this deadline (e.g. if you cannot retrieve necessary files from your laboratory, etc.), please let us know and we can work with you to determine a suitable revision period.

Please contact the journal office with any questions, cellbio@rockefeller.edu or call (212) 327-8588.

Thank you for this interesting contribution, we look forward to publishing your paper in Journal of Cell Biology.

Sincerely,

Melissa Gardner Monitoring Editor

Andrea L. Marat Senior Scientific Editor -----

Reviewer #1 (Comments to the Authors (Required)):

The authors have fully addressed my comments.